Author's response to reviews

Title: The diagnostic value of endoscopy and Helicobacter pylori tests for peptic ulcer patients in late post-treatment setting

Authors:

Heidi-Ingrid Maaroos (Heidi-Ingrid.Maaroos@ut.ee)
Helena Andreson (Helena.Andreson@ut.ee)
Krista Loivukene (Krista.Loivuke@kliinikum.ee)
Pirje Hutt (Pirje.Hutt@ut.ee)
Helgi Kolk (Helgi.Kolk@kliinikum.ee)
Ingrid Kull (Ingrid.Kull@kliinikum.ee)
Katrin Labotkin (Katrin.Labotkin@kliinikum.ee)
Marika Mikelsaar (Marika.Mikelsaar@ut.ee)

Version: 3 Date: 23 October 2004

Author's response to reviews:

Cover letter

Title: The diagnostic value of endoscopy and Helicobacter pylori tests for peptic ulcer patients in late post-treatment setting

Authors:

Heidi-Ingrid Maaroos: Heidi-Ingrid.Maaroos@ut.ee
Helena Andreson: Helena.Andreson@ut.ee
Krista Loivukene: Krista.Loivuke@kliinikum.ee
Pirje Hutt: Pirje.Hutt@ut.ee
Helgi Kolk: Helgi.Kolk@kliinikum.ee
Ingrid Kull: Ingrid.Kull@kliinikum.ee
Katrin Labotkin: Katrin.Labotkin@kliinikum.ee
Marika Mikelsaar: Marika.Mikelsaar@ut.ee

Version: 3 Date: 23 Oct 2004

Dear editor,

Thank you for accepting our manuscript.

We made the suggested formatting changes on pages 1, 5-7 and 14, concerning affiliations, section sub-headings and competing interests, respectively.